Author’s response to reviews

Title: Distinct molecular etiologies of male and female hepatocellular carcinoma

Authors:
Heini Maaret Natri (hnatri@asu.edu)
Melissa A Wilson (mwsayres@asu.edu)
Kenneth H Buetow (kenneth.buetow@asu.edu)

Version: 2 Date: 11 Sep 2019

Author’s response to reviews:

We have revised the manuscript according to the editors comments:

"Please include a correctly formatted abstract as described in our instructions to authors."

We have revised the abstract to match the journal guidelines.

"Please include a Keywords section to follow the Abstract. Three to ten keywords representing the main content of the article should be included."

We have included keywords.

"Please revise your ‘Declaration’ section to be in the following order with these headings:
Declarations
- Ethics approval and consent to participate
- Consent for publication
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information (optional)"

We have reordered and added the missing information in the declaration section.

"In the Funding section of the Declarations please indicate the role of the funding body in the design of
the study and collection, analysis, and interpretation of data and in writing the manuscript. If no specific funding was received for this study, please clearly indicate this in the Funding section.”

We have added a statement in the funding section.